Author's response to reviews

Title: Cost-effectiveness of introducing a rotavirus vaccine in developing countries: the case of Mexico

Authors:

Atanacio Valencia-Mendoza (avalencia@correo.insp.mx)
Stefano M Bertozzi (sbertozzi@correo.insp.mx)
Juan Pablo Gutierrez (jpgutier@correo.insp.mx)
Robbin Itzler (robbin_itzler@merck.com)

Version: 3 Date: 21 July 2008

Author's response to reviews:

Dr. Melissa Norton
Editor-in-Chief
BMC Infectious Diseases

Dear Dr. Norton:

Attached please find the second version of the revised manuscript entitled “Cost-effectiveness of Introducing a Rotavirus Vaccine in Developing Countries: the Case of Mexico,” MS: 1677367371726492, which we hereby resubmit for BMC Infectious Diseases.

Since the reviewers are already satisfied with the actual version of the manuscript, the changes made in this version were not others than those formatting changes suggested by the BioMed Central Editorial Production Team.

Sincerely,

Atanacio Valencia-Mendoza